

THE UNIVERSITY of EDINBURGH

# Edinburgh Research Explorer

# Alternatively activated macrophages promotes necrosis resolution following acute liver injury

# Citation for published version:

Starkey Lewis, PJ, Campana, L, Aleksieva, N, Cartwright, J, Mackinnon, A, O'Duibhir, E, Kendall, T, Vermeren, M, Thomson, AW, Gadd, V, Dwyer, B, Aird, R, Man, T-Y, Rossi, A, Forrester, L, Park, BK & Forbes, S 2020, 'Alternatively activated macrophages promotes necrosis resolution following acute liver injury', *Journal of Hepatology*. https://doi.org/10.1016/j.jhep.2020.02.031

# **Digital Object Identifier (DOI):**

10.1016/j.jhep.2020.02.031

## Link:

Link to publication record in Edinburgh Research Explorer

## **Document Version:**

Version created as part of publication process; publisher's layout; not normally made publicly available

Published In: Journal of Hepatology

## **Publisher Rights Statement:**

This is a PDF file of an article that has undergone enhancements after acceptance, such as the addition of a cover page and metadata, and formatting for readability, but it is not yet the definitive version of record. This version will undergo additional copyediting, typesetting and review before it is published in its final form, but we are providing this version to give early visibility of the article. Please note that, during the production process, errors may be discovered which could affect the content, and all legaldisclaimers that apply to the journal pertain.© 2020 Published by Elsevier B.V. on behalf of European Association for the Study of the Liver.

## **General rights**

Copyright for the publications made accessible via the Edinburgh Research Explorer is retained by the author(s) and / or other copyright owners and it is a condition of accessing these publications that users recognise and abide by the legal requirements associated with these rights.

## Take down policy

The University of Édinburgh has made every reasonable effort to ensure that Edinburgh Research Explorer content complies with UK legislation. If you believe that the public display of this file breaches copyright please contact openaccess@ed.ac.uk providing details, and we will remove access to the work immediately and investigate your claim.



Alternatively activated macrophages promote resolution of necrosis following acute liver injury

Philip Starkey Lewis, Lara Campana, Niya Aleksieva, Jennifer Ann Cartwright, Alison Mackinnon, Eoghan O'Duibhir, Timothy Kendall, Matthieu Vermeren, Adrian Thomson, Victoria Gadd, Benjamin Dwyer, Rhona Aird, Tak-Yung Man, Adriano Giorgio Rossi, Lesley Forrester, Brian Kevin Park, Stuart John Forbes



JOURNAL OF HEPATOLOGY

PII: S0168-8278(20)30133-1

DOI: https://doi.org/10.1016/j.jhep.2020.02.031

Reference: JHEPAT 7648

- To appear in: Journal of Hepatology
- Received Date: 17 September 2019
- Revised Date: 19 February 2020

Accepted Date: 24 February 2020

Please cite this article as: Lewis PS, Campana L, Aleksieva N, Cartwright JA, Mackinnon A, O'Duibhir E, Kendall T, Vermeren M, Thomson A, Gadd V, Dwyer B, Aird R, Man TY, Rossi AG, Forrester L, Park BK, Forbes SJ, Alternatively activated macrophages promote resolution of necrosis following acute liver injury, *Journal of Hepatology* (2020), doi: https://doi.org/10.1016/j.jhep.2020.02.031.

This is a PDF file of an article that has undergone enhancements after acceptance, such as the addition of a cover page and metadata, and formatting for readability, but it is not yet the definitive version of record. This version will undergo additional copyediting, typesetting and review before it is published in its final form, but we are providing this version to give early visibility of the article. Please note that, during the production process, errors may be discovered which could affect the content, and all legal disclaimers that apply to the journal pertain.

© 2020 Published by Elsevier B.V. on behalf of European Association for the Study of the Liver.

**Title:** Alternatively activated macrophages promote resolution of necrosis following acute liver injury

**Authors:** Philip Starkey Lewis<sup>1,4</sup>, Lara Campana<sup>1,2</sup>, Niya Aleksieva<sup>1</sup>, Jennifer Ann Cartwright<sup>2</sup>, Alison Mackinnon<sup>2</sup>, Eoghan O'Duibhir<sup>1</sup>, Timothy Kendall<sup>2,3</sup>, Matthieu Vermeren<sup>1</sup>, Adrian Thomson<sup>5</sup>, Victoria Gadd<sup>1</sup>, Benjamin Dwyer<sup>1</sup>, Rhona Aird<sup>1</sup>, Tak-Yung Man<sup>1</sup>, Adriano Giorgio Rossi<sup>2</sup>, Lesley Forrester<sup>1</sup>, Brian Kevin Park<sup>4,6</sup> & Stuart John Forbes\*<sup>1,4</sup>

Affiliations: <sup>1</sup>MRC Centre for Regenerative Medicine, Little France Drive, University of Edinburgh, Edinburgh, United Kingdom; <sup>2</sup>Centre for Inflammation Research, Queen's Medical Research Institute, University of Edinburgh, Edinburgh, United Kingdom; <sup>3</sup>Edinburgh Pathology, University of Edinburgh, Edinburgh, United Kingdom; <sup>4</sup>UK Regenerative Medicine Platform Safety and Efficacy Hub, University of Liverpool, Liverpool, United Kingdom; <sup>5</sup>Edinburgh Preclinical Imaging, BHF Centre for Cardiovascular Science, University of Edinburgh, Edinburgh, Edinburgh, Edinburgh, Edinburgh, Edinburgh, University of Liverpool, Liverpool, Liverpool, Liverpool, Liverpool, Liverpool, Liverpool, Liverpool, United Kingdom;

**Corresponding author:** Prof. Stuart John Forbes, Director of MRC Centre for Regenerative Medicine, Little France Drive, University of Edinburgh, Edinburgh bioQuarter, Edinburgh, United Kingdom, EH16 4UU. Telephone: +44(0)1316519636, Fax: +44(0)1316519501 E-mail: stuart.forbes@ed.ac.uk

Keywords: Acetaminophen; Liver Regeneration; Macrophages; Necrosis; Phagocytosis

Electronic word count: 6,351

Number of main figures and tables: 5

**Conflict of interest statement:** P.S.L., L.F., and S.J.F. have patents pending, entitled 'Macrophage-based therapy' in national territories of USA, Europe, Japan, China and Australia. These patents have been derived from PCT/GB2017/052769 filed 18/09/2017 and claim priority from UK application 1615923.8 filed 19/09/2016. Both of the original patents have now been abandoned because the original UK patent and PCT patent are no longer live and have now been replaced by the national patents.

**Financial support statement:** P.S.L. was supported by grant funding from the UK Regenerative Medicine Platform (UKRMP) Safety and Efficacy Hub (grant ref MR/K026739/1), and Stem Cell Niche Hub. Human macrophage studies were supported by an MRC Confidence in Concept award (MRC/CIC5/06), with research collaboration funding from Syncona Ltd. (funder reference: MED2589). L.C. was supported by MRC grant (MR/J010766/1). T.J.K. was supported by Wellcome Trust Intermediate Clinical Fellowship (095898/Z/11/Z). J.C. was supported by Wellcome Trust grant (10896/Z/15/Z). N.A. was supported by Wellcome Trust grant (203925/Z/16/A). Funding sources had no involvement in study design, data collection, data analysis and interpretation, writing of the report, nor decision to publish.

**Authors contributions:** S.J.F. and P.S.L. conceived the study concept and experimental design. P.S.L., L.C., J.C., A.M., A.T., V.G., R.A., and T-Y.M. performed experiments. P.S.L., L.C., N.A., J.C., and B.D, collected data and performed statistical analysis. T.K., E.O., and M.V. provided analytical and imaging support. A.R., L.F., and K.P. provided intellectual input and supervision. P.S.L., L.C., and S.J.F. drafted the manuscript. All authors reviewed and edited the manuscript.

# Abstract.

Background & Aim: Acute liver injury (ALI) can occur if a significant acetaminophen (APAP) overdose presents too late for n-acetylcysteine treatment, which risks deterioration into acute liver failure, systemic inflammation, and death. Macrophages influence the progression and resolution of ALI due to their innate immunological function and paracrine activity. Syngeneic primary bone-marrow derived macrophages (BMDMs) were tested as a cell-based therapy in a mouse model of APAP-ALI. Methods: Several phenotypically-distinct BMDM populations were delivered intravenously to APAP-ALI mice when hepatic necrosis was established, and then evaluated based on their effects on injury, inflammation, immunity, and regeneration. In vivo phagocytosis assays were used to interrogate the phenotype and function of alternativelyactivated BMDMs (AAMs) post-injection. Finally, primary human AAMs sourced from healthy volunteers were evaluated in immunocompetent APAP-ALI mice. Results: BMDMs rapidly localised in liver and spleen within four hours of administration. Injection of AAMs specifically reduced hepatocellular necrosis, HMGB1 translocation, and infiltrating neutrophils following APAP-ALI. AAM delivery also stimulated proliferation in hepatocytes and endothelium, and reduced levels of several circulating proinflammatory cytokines within 24 hours. AAMs displayed a high phagocytic activity both in vitro and in injured liver tissue post-injection. Crosstalk with the host innate immune system was evidenced by reduced infiltrating host Ly6C<sup>hi</sup> macrophages in AAM-treated mice. Importantly, therapeutic efficacy was partially recapitulated using clinical-grade primary human AAMs in immunocompetent APAP-ALI mice underscoring translational potential. Conclusion: We identify that AAMs have value as a cell-based therapy in an experimental model of APAP-ALI. Human AAMs warrant further evaluation as a potential cell-based therapy for APAP overdose patients with established liver injury.

Lay summary: After an overdose of acetaminophen (paracetamol), some patients present to hospital too late for the current antidote (n-acetylcysteine) to be effective. We tested whether macrophages, an injury-responsive leukocyte that can scavenge dead/dying cells, could serve as a cell-based therapy in an experimental model of acetaminophen overdose. Injection of , μry an as a potential c alternatively-activated macrophages rapidly reduced liver injury and reduced several mediators of inflammation. Macrophages show promise to serve as a potential cell-based therapy for acute liver injury.

4

# Introduction

Acetaminophen (paracetamol, APAP) overdose is a common cause of acute liver injury (ALI) in the clinic and is the leading cause of acute liver failure (ALF) in the United States.<sup>[1-3]</sup> APAP also serves as a paradigm hepatotoxin for pre-clinical studies and the molecular mechanisms that underpin APAP hepatotoxicity are relatively well understood. Therapeutic management of APAP-ALI is primarily limited to n-acetylcysteine (NAC) therapy, which serves as an effective antidote. However, NAC efficacy is substantially diminished in patients who present late after APAP ingestion (i.e. longer than 10 hours).<sup>[4]</sup> Liver transplantation may be required in patients who subsequently develop ALF. However, due to the shortages of suitable donor tissue, and associated life-long immunosuppression, liver transplantation is not an ideal therapeutic intervention. Therefore, novel therapies to prevent liver injury progressing to acute liver failure are urgently sought.

APAP-induced acute liver injury (APAP-ALI) is characterised by fulminant hepatocyte necrosis. Without immediate NAC-treatment, substantial liver injury can progress into ALF associated with systemic inflammatory response syndrome (SIRS) characterised by immune activation and encephalopathy, conferring a high risk of multi-organ failure and death.<sup>[5]</sup> Recent work has shown that liver-resident macrophages (Kupffer Cells, KCs), which provide hepatic innate immunity (e.g. against gut-derived pathogens), are substantially reduced during APAP-ALI leading to a transient immunological perturbation in the liver.<sup>[6-8]</sup> Patients with ALF frequently develop enteric bacterial and fungal infections (typically *Escherichia coli* and *Candida*), which are often associated with fatal outcomes.<sup>[9,10]</sup> The role of macrophages in the pathology of APAP-ALI has raised a conflicting literature. Chemical pre-treatment to ablate KCs in rats before APAP-ALI showed protective effects suggesting macrophages can exacerbate liver injury

through release of proinflammatory mediators.<sup>[11-13]</sup> However, subsequent studies found that macrophages are absolutely required for appropriate tissue repair and angiogenesis following APAP-ALI.<sup>[14-16]</sup> Both KCs and infiltrating macrophages acquire distinct but restorative phenotypes that are required for the timely resolution of APAP-ALI.<sup>[6]</sup> Recent work showed that human liver macrophages in ALF patients also acquire hepatoprotective phenotypes characterised by high phagocytic function and expression of clearance receptors, e.g. Mertk.<sup>[17]</sup> The clearance of apoptotic and necrotic cells are orchestrated primarily by macrophages, which are fundamentally required to resolve inflammation and injury effectively.<sup>[18,19]</sup> Therefore, we hypothesised that injection of primary macrophages may serve as a cell therapy for APAP-ALI in order to facilitate clearance of necrotic material, reduce local and systemic inflammation, and promote liver regeneration. Primary macrophages can be differentiated from bone-marrow precursors in vitro to yield a highly-enriched population of functional bone-marrow derived macrophages (BMDMs). Injection of BMDMs has previously been shown to ameliorate liver fibrosis in chronic liver injury models.<sup>[20]</sup> Clinical-grade autologous human monocyte-derived macrophages (hMDMs) have recently been found safe in cirrhotic patients, with phase 2 efficacy trials in progress (ISRCTN 10368050).<sup>[21-23]</sup>

Here, we have tested phenotypically-distinct BMDM populations in a murine model of APAP-ALI. We report that administration of alternatively-activated macrophages (AAMs) reduces necrotic area, reduces several proinflammatory cytokines in tissue and serum, and stimulates hepatocellular proliferation. Importantly from a translational aspect, clinical-grade human AAMs (hAAMs) recapitulated some efficacy readouts in immunocompetent APAP-ALI mice. Our study identifies a potential cell therapy <u>for established APAP-ALI</u> with clinical applicability for a patient group with limited therapeutic options.

# **Materials and Methods**

# **BMDM production**

BMDMs were prepared as previous, with minor modifications.<sup>[20]</sup> Mouse bone marrow (BM) was flushed from femurs and tibias of healthy C57BL/6JOlaHsd male mice (8-10 weeks old, Envigo). BM suspensions were filtered (70 μm) into DMEM:F12 (1:1) cell culture media (Gibco) supplemented with fetal bovine serum (FBS, 10 %), L-glutamine (2 mM), penicillin/streptomycin (100 U/mL, 100 µg/mL), and murine recombinant CSF1 (40 ng/mL; Peprotech). BM preps were incubated in ultra-low attachment flasks (Corning Inc.) for 7 days (37 °C, 5 % CO<sub>2</sub>), with additional feeds on days 3 and 5 (20 ng/mL CSF1, in a 50 % media change) to produce BMDMs. BMDMs were polarised with recombinant factors overnight to generate classically-activated macrophages (CAMs; with LPS, Sigma-Aldrich, 50 ng/mL; and IFNγ, 20 ng/mL, Peprotech), AAMs (with IL-4 and IL-13; 20 ng/mL each, Peprotech), or deactivated macrophages (DAMs; with IL-10, 10 ng/mL, Peprotech). In some experiments, BMDMs were labelled with CellTrace CFSE (ThermoFisher) *in vitro*, following the manufacturer's instructions.

# **hMDM production**

For phagocytosis assays, non-GMP hMDMs were differentiated from cryopreserved primary CD14+ monocytes using serum-containing Iscove's Modified Eagle's Medium essentially as described using human recombinant CSF-1 (100 ng/mL, Peprotech).<sup>[21]</sup> hCAMs and hAAMs were polarised from hMDMs by overnight stimulation with LPS/hIFNγ (50/20 ng/mL), and hIL-4/IL-13 (20 ng/mL each) respectively. For clinical-grade hMDMs, we utilised a serum-free

GMP-compliant process as described.<sup>[22]</sup> Peripheral blood mononuclear cells (PBMCs) were centrifuged and collected from healthy volunteer buffy coats using Ficoll-paque 1.077 (GE Healthcare). CD14+ cells were isolated from PBMCs using CliniMACS CD14 MicroBeads (Miltenyi-Biotec) on LS separation columns (Miltenyi Biotec). CD14+ cells were cultured (37 °C, 5 % CO<sub>2</sub>) for 7 days in TexMACS GMP media (Miltenyi Biotec), supplemented with GMP-grade human recombinant CSF-1 (100 ng/mL, R&D Bio-Techne) with an additional feed at day 3 to generate hMDMs. Clinical-grade hAAMs were generated from hMDMs using human recombinant cytokines (R&D Bio-Techne) as above. Successful hMDM differentiation was confirmed using flow cytometry to demonstrate a minimum five-fold MFI increase on 25F9 and CD206 compared to initial CD14+ cells.

# APAP-ALI and macrophage administration.

Eight week old male C57BL6/JOIaHsd mice housed in open top cages were fasted for 14 hours. All mice received a single APAP injection (350 - 500 mg/kg, i.p., Sigma Aldrich) in warm sterile saline (PanReac Applichem). Standard chow and wet mash was returned to mice 20 min post-APAP administration. Macrophages were resuspended in PBS (Sigma) and administered (1- $5x10^{6}$  cells, i.v., 100-200 µL) to APAP-ALI mice at 16 hours. PBS alone (100-200 µL, i.v.) served as vehicle control for macrophage treatment. After macrophage/vehicle treatment, all mice were transferred to a warming cabinet (28 c). One hour before cull, mice were pulsed with 5-bromo-2'-deoxyuridine (BrdU, 1 mg in sterile saline, i.p., Sigma Aldrich) to label proliferating cells. Mice were humanely culled and whole blood was collected via cardiac puncture. For immunocompromised mice, the same methodology was used except fasted 12-

week old male NOD.Cg-Prkdc<sup>scid</sup> Il2rg<sup>tm1WjI</sup>/SzJ (NSGs), housed in individually-ventilated cages, received 250 mg/kg APAP (i.p.). Substances were administered to NSGs aseptically.

# Statistics.

Statistical analysis was performed in Prism 8.2 (GraphPad Software). All data are presented as individual scatter plots to show each experimental unit (e.g. individual mice) unless otherwise stated. To test two groups, an unpaired two-way t-test or Mann-Whitney U-test was performed on parametric and non-parametric datasets respectively. To test more than two parametric groups, a one-way ANOVA, two-way ANOVA (with Dunnett's multiple comparison test), or mixed-effects model (with Sidak's multiple comparison's test) was performed. To test more than two non-parametric groups, a Kruskal-Wallis test (with Dunn's multiple comparison test) was performed. Shapiro-Wilk test determined normality. Sample size was determined based on power calculation ( $\alpha = 0.05$ , desired power = 0.8) or from investigator experience. *P* < 0.05 was considered statistically significant.

# **Study Approval.**

All animal experiments undertaken in accordance with criteria outlined in a license granted under the Animals (Scientific Procedures) Act 1986 and approved by the University of Edinburgh Animal Ethics Committee. Use of human material was granted by the South East Scotland Research Ethics Committee 02, and use of buffy coats was covered by Scottish National Blood Transfusion Service (SNBTS). Buffy coats from informed consenting healthy volunteers were obtained in collaboration with SNBTS Blood Donor Centre, Edinburgh, United Kingdom, under SNBTS Sample Governance 16-09. For further information on flow cytometry, imaging, and other materials and methods, including CTAT table, please refer to the supplementary information.

#### Results

<u>AAM-administration reduces hepatic necrosis and stimulates hepatocellular proliferation</u> <u>following APAP-ALI</u>

First, an appropriate delivery route for injecting BMDMs was identified. We tested whether BMDMs could be injected intravenously (i.v.) to rapidly deliver BMDMs to the liver, since this route is fast, non-invasive, and clinically-applicable in the setting of APAP-ALI. In vivo and ex vivo imaging techniques demonstrated a linear accumulation of BMDMs in the liver and spleen over the first four hours post-injection (Fig. S1). Next, we tested the efficacy of four different macrophage populations (Naïve BMDMs, CAMs, AAMs, or DAMs) as a cell-based therapy for APAP-ALI (Fig. 1A; expanded schematic in Fig. S2A). All macrophage populations showed high enrichment for CD11b and F4/80 or CSF1R, and expressed typical markers associated with their activation status (Fig. S2D-G). Each macrophage population (or PBS alone) was administered (1x10<sup>6</sup> cells, 100  $\mu$ L, i.v.) to mice with APAP-ALI at 16 hours when ALI is established (Fig. 1B). Macrophage administration was well tolerated showing no serum chemistry changes in healthy mice (Fig. S3), and no impact on hematology parameters in APAP-ALI mice (Fig. S4). Serum transaminase activity was moderately lower in all BMDM-treated groups compared to PBS-treated controls in APAP-ALI mice, but was not significantly different (Fig. 1C). Serum transaminases have a circulating half-life of several hours,<sup>[24]</sup> therefore necrotic area was quantified from haematoxylin and eosin stained liver sections for a direct histological measure. AAM-treated mice showed a specific 60 % reduction in necrotic area compared to PBS-treated controls in APAP-ALI mice (Fig. 1D). In parallel experiments, AAM-treatment reduced necrotic area when injected at 6 hours post-APAP at 400 mg/kg (Fig. S5). Neutrophils provide a major source of reactive oxygen species (ROS) that contribute to early injury, although this has been disputed.<sup>[25-28]</sup> Macrophages have been implicated in the removal of neutrophils during inflammation.<sup>[29]</sup> Neutrophils (detected by Ly6G immunostaining) were 52 % lower in necrotic areas specifically in AAM-treated mice compared to PBS-treated controls (Fig. 1E). Next, HMGB1, a damage associated molecular pattern (DAMP), was immunostained in liver sections. Translocation of nuclear HMGB1 into the cytoplasm is an early critical step for its extracellular release.<sup>[30]</sup> Peri-necrotic hepatocytes in APAP-ALI showed HMGB1 cytosolic localisation, but the frequency was 66 % lower in AAM-treated mice (Fig. 1F). To measure liver regeneration, all mice received BrdU (1 mg, i.p.) one hour before cull to label proliferating cells at sacrifice. BrdU-incorporation in liver tissue was higher in mice treated with CAMs (8.5-fold) and AAMs (8.4-fold) compared to PBS-treated controls (Fig. 1G). These data indicated that AAMs provided the greatest therapeutic response. Therefore further experiments focused on AAM-treatment, with naïve BMDMs serving as a cell-treatment reference group, and PBStreatment serving as the vehicle-control group. Dual immunofluorescence (IF) staining revealed proliferating cells after AAM-treatment included both hepatocytes and endothelial cells evidenced by BrdU co-localisation with HNF4a and ERG respectively (Fig. 2H, I). Immunostains for isotype antibody controls are provided in Fig. S6. In separate studies, AAMs were tested for efficacy at 500 mg/kg APAP where mortality is expected. Survival experiments are not in compliance with Home Office regulations in the United Kingdom, therefore we established a phenotypic clinical scoring system (see Table S1). Pre-defined thresholds triggered

a humane cull in order to prevent mice exceeding severity limits. It was necessary to deliver AAMs at four hours post-APAP to test efficacy because deaths can occur within 8 hours at high APAP doses in mice. We observed no benefit in mice receiving AAMs ( $1x10^6$  cells, i.v.) in this experiment (Fig. S7) suggesting these cells are more efficacious when injected at 16 hours.

# <u>AAM-administration reduces several inflammatory cytokines in serum and liver tissue following</u> <u>APAP-ALI</u>

To test the effect on inflammation after AAM-treatment, a panel of cytokines were quantified in serum and liver homogenates obtained from APAP-ALI mice. <u>Consistently, AAM-treatment</u> specifically reduced several serum pro-inflammatory cytokines including IFN- $\gamma$  (82 %), IL-12p70 (73 %), and IL-6 (75 %) versus PBS-treated controls (Fig. 2A). Treatment with both AAMs and naïve BMDMs reduced serum CXCL1 levels. Serum TNF $\alpha$  was 39 % lower after BMDM-treatment suggesting that naïve BMDMs may exert subtle effects on circulating cytokines. Importantly, hepatic IL-6 levels were 34 % lower specifically in AAM-treated liver homogenates compared to PBS-treated controls (Fig. 2B). Further, we measured the expression of a panel of genes associated with inflammation in whole liver tissue. We observed 3.3-fold higher *Tgfb* and lower *116* expression levels were observed in both AAM- and naïve BMDM-treated mice (Fig. 2C). Expression of other inflammatory-associated genes such as *Cxcl1* and *Ccl5* were lower on average in AAM-treated liver but did not reach statistical significance. To better understand the trajectory of recovery following AAM-treatment, a longitudinal experiment was performed in APAP-ALI mice by performing serial blood microsampling up to 84 hours

(study design: <u>Fig. 2D</u>). Plasma levels of ALT, AST, and the hepatocyte-specific microRNA, miR-122, showed significantly bigger reductions after AAM-treatment compared to PBS-treated controls suggesting improved recovery (<u>Fig. 2E-G</u>).

# Murine AAMs are primarily Ly6C<sup>lo</sup> and highly phagocytic in APAP-ALI liver tissue postinjection

A series of experiments was performed to assess the phenotype and function of AAMs in vitro and in vivo. Gene expression analysis confirmed polarisation occurred in vitro for all BMDM populations, e.g. Nos2 was 775-fold higher in CAMs, Retnla was 4850-fold higher in AAMs, whilst *ll10* and *Ly6C* expression was higher in DAMs and CAMs respectively (Fig. 3A; further data in Fig. S2B). Flow cytometry revealed that all BMDM subsets were approximately 80 % Ly6C<sup>lo</sup> in vitro (Fig. 3B). To investigate AAM phagocytosis in vivo, CFSE-labelled AAMs were administered (5x10<sup>6</sup>, i.v.) at 16 hours, three hours before administration of PKH26PCL (PKH, i.v.), a fluorescent probe that specifically labels cells performing phagocytosis (Fig. 3C). A higher number of AAMs were injected in this experiment to improve assay sensitivity. Serum ALT activity was significantly lower in AAM-treated mice compared to PBS-treated controls (Fig. 3D). A flow cytometry gating strategy was used to analyse myeloid cells in liver digests (Fig. S8). CFSE+ AAMs represented 0.9 % and 0.5 % of the CD11b+ population in liver digests and whole blood respectively in APAP-ALI mice (Fig. 3E). Low expression of Ly6C, a cell surface myeloid marker, has been associated with a restorative macrophage phenotype during liver disease.<sup>[31]</sup> Injected AAMs remained approximately 80 % Ly6C<sup>lo</sup> at 36 hours in APAP-ALI mice (Fig. 3F). Furthermore, AAMs were highly phagocytic showing high PKH-uptake in

Ly6C<sup>hi</sup> (69 % positive) and Ly6C<sup>lo</sup> (99 % positive) AAMs (Fig. 3G). We observed a 5 % reduction in infiltrating endogenous Ly6C<sup>hi</sup> macrophages in AAM-treated mice (Fig. 3H). Furthermore, PKH-uptake in Ly6C<sup>hi</sup> infiltrating endogenous macrophages increased from 8.3 % in PBS-treated controls to 12 % in AAM-treated mice, whilst PKH-uptake was equivalent in Ly6C<sup>lo</sup> infiltrating macrophages (Fig. 3I) suggesting crosstalk exists between AAMs and the host innate immune response. Finally, to test if AAMs alter their phenotype post-injection, we performed a low-density PCR array on FACS-sorted AAMs from healthy and APAP-ALI liver digests. We found only four genes (*II10*, 15.3-fold; *C4b*, 8.8-fold, *Tlr4* 6.1-fold; and *B2m*, 6.2-fold) from 84 tested genes were downregulated in AAMs after injection into APAP-ALI mice compared to injection into healthy mice (Fig. S9A-D and Table S2). These data suggest that injected AAMs are highly phagocytic *in situ*, and largely retain their phenotype post-injection.

# Murine AAMs are highly phagocytic in vitro

To further understand the phenotype and function of AAMs, we performed a series of *in vitro* phagocytosis experiments. The propensity of BMDM populations to phagocytose apoptotic material was assessed by flow cytometry after incubating BMDMs with (5-(and-6)-(((4-chloromethyl)benzoyl)amino)tetramethylrhodamine)- (CMTMR-) labelled apoptotic thymocytes using a gating strategy (Fig. S10). The mean fluorescence intensity (MFI) was higher in AAMs at 30 min versus other groups (21 % increase vs naïve, 75 % increase vs CAMs) and at 60 mins versus CAMs (64 % increase; Fig. 4A). Furthermore, AAMs had a significantly lower percentage of Ly6C<sup>hi</sup> cells at 30 min (22 % lower vs naïve, 72 % vs CAMs), 60 min (44 % lower vs naïve, 52 % vs CAMs), and 120 min (61 % lower vs naïve, 72 % vs CAMs) suggesting AAMs

possess a sustained phenotype throughout phagocytosis (Fig. 4B). In parallel, real-time live imaging assays showed AAMs could phagocytose pHrodo-bioparticles faster and to a greater extent than other BMDM populations (Fig. 4C, D, and video S1). Similarly, hAAMs produced from cryopreserved human CD14+ cells also showed similar enhanced phagocytic performance *in vitro* (Fig. S11). To test if AAMs could phagocytose necrotic hepatocytes *in vivo*, CFSE+ AAMs (5x10<sup>6</sup>, i.v.) were injected into APAP-treated R26RLSL tdTomato mice two weeks after viral-Cre delivery to induce tdTomato in hepatocytes (Fig. 4E). Injected AAMs localised throughout the liver parenchyma, with many surrounding necrotic areas (Fig. S12A). Labelled mice developed APAP-ALI characterised by a loss of tdTomato hepatocytes around the central vein and raised serum ALT activity (Fig. S12A, B). Confocal microscopy revealed that perinecrotic AAMs contained multiple intracellular vesicles containing faint fluorescent punctate tdTomato-positive material consistent with surrounding hepatocellular debris (Fig. 4F). In contrast, no tdTomato material was visible in sinusoidal AAMs located distal to necrotic areas in the same tissue.

# <u>Administration of human AAMs reduces necrosis and stimulates hepatocyte proliferation</u> <u>following APAP-ALI in immunocompetent mice</u>

A fully-defined serum-free protocol to generate clinical-grade human monocyte-derived macrophages (hMDMs) has recently been described.<sup>[22]</sup> Here, we modified this protocol to generate clinical-grade hAAMs for evaluation in the APAP-ALI model (Fig. 5A; expanded schematic Fig. S13A). Isolation of CD14+ cells using CliniMACS microbeads from human PBMCs provided significant CD14+ CD45+ enrichment (Fig. 5B). CD14+ cells were cultured

with GMP-grade recombinant hCSF-1 to generate hMDMs. Flow cytometry revealed hMDMs upregulated human macrophage maturity markers 25F9 and CD206 after 7 days (Fig. 5C). hMDMs were stimulated with hIL-4/-13 in vitro for 24 hours to generate hAAMs, exhibiting higher CD206, CD163 and CD169 surface levels (Fig. 5D). hAAMs had higher MRC1, DC-SIGN, SCARB1 gene expression, and lower NOS2 compared to volunteer-matched hMDMs (Fig. S13B). Firstly, hAAMs were tested in an immunocompromised APAP-ALI model  $(1 \times 10^6, i.v., i.v.)$ NSG mice, 250 mg/kg). NSG mice could only tolerate mild centrilobular necrosis, therefore a lower APAP dose was used (250 mg/kg). However, hAAMs were not efficacious in the NSG model showing no change in necrosis or liver injury markers (Fig. S14). Therefore, we re-tested hAAMs (5x10<sup>6</sup>, i.v.) in an immunocompetent strain (C57BL/6J) with APAP-ALI. Firstly, healthy C57BL/6J mice tolerated hMDM-injection well showing no changes in body weight, blood circulating monocytes or blood neutrophils at 24 hours or 7 days post-injection with no phenotypic evidence of severe acute rejection (Fig. S15). Human cells were expected to be rapidly cleared in immunocompetent mice. Nevertheless, some FITC-positive cells were detected in liver and spleen of APAP-ALI mice treated with CFSE-labelled AAMs suggesting transient hepatic localisation is possible (Fig. 5E). Histological evaluation showed 32 % reduced necrosis specifically in hAAMs-treated mice (Fig. 5F). Both hMDM- and hAAM-treatment led to reduced weight loss, which was not related to any changes in liver mass indicated by consistent liver/body weight ratios (Fig. 5G). Similar to murine AAM-treatment, serum ALT activity showed a strong reduced trend in hAAM-treated mice compared to PBS-treated controls, but did not reach statistical significance (Serum ALT, P = 0.056; Fig. 5H). We observed no changes in serum cytokines, including murine IL-6, in hAAM-treated mice versus PBS-controls, although murine IL-10/IL-12 ratio displayed a higher trend in hAAM-treated mice (P=0.07, Fig. 5H).

<u>Importantly</u>, an increase in BrdU-incorporation was observed in HNF4 $\alpha$ -positive nuclei specifically in hAAM-treated liver tissues (<u>Fig. 51</u>) suggesting improved hepatocyte proliferation. These data suggest that hAAMs can recapitulate some efficacy readouts of murine <u>AAMs in experimental APAP-ALI.</u>

# Discussion

NAC is the primary treatment option for APAP overdoses and acts through boosting the antioxidant capacity of the liver to prevent liver injury occurring. NAC efficacy is diminished in patients who present to hospital late (i.e. 10-12 hours after APAP ingestion) when liver injury may already be established. Therefore, novel therapeutic strategies are required to treat late-presenting APAP-ALI patients with established liver injury. Here, we demonstrate a novel cell-based immunotherapeutic approach to promote necrosis resolution, reduce systemic inflammation, and expedite liver repair in experimental APAP-ALI. Recent work has shown that KCs, highly phagocytic liver-resident macrophages, are depleted during APAP-ALI impairing hepatic innate immunity.<sup>[6]</sup> Large numbers of monocytes rapidly infiltrate the liver thereafter and differentiate into monocyte-derived macrophages *in situ*, but they are inflammatory (e.g. Ly6C<sup>hi</sup>) and poorly phagocytic, at least initially.<sup>[6,32,33]</sup> Therefore, we hypothesised that timely intervention with polarised macrophages may hold therapeutic value as a cell-based therapy for ALI to facilitate necrosis resolution and promote liver regeneration. Macrophages have previously been evaluated as an experimental cell-based therapy for other diseases.<sup>[34,35]</sup> including liver fibrosis,<sup>[20]</sup> but have not yet been tested in the setting of ALI.

Hepatocyte necrosis during APAP-ALI is the priming event to activate the innate immune system, through release of proinflammatory cytokines and danger associated molecular patterns (DAMPs), e.g. HMGB1.<sup>[36]</sup> We observed that AAM-treatment specifically reduced necrotic area in APAP-ALI mice within 24 hours. AAMs also led to reductions in infiltrating neutrophils, cytosolic HMGB1 translocation (in peri-necrotic hepatocytes), and attenuated several serum proinflammatory cytokines suggesting AAMs exert an anti-inflammatory effect. This has potential clinical importance because systemic inflammation (*via* uncontrolled activation of the innate immune system) is a feature of systemic inflammatory response syndrome (SIRS) - a key determinant of clinical outcome in APAP-ALF.<sup>[5]</sup> Furthermore, AAM-treatment also led to improved hepatic proliferation, particularly in hepatocytes and endothelium. Hepatic proliferation and revascularisation are recognised as important features of liver regeneration following injury.<sup>[37-39]</sup>

AAM efficacy was associated with a highly phagocytic phenotype, therefore we posited that AAM delivery may augment clearance of necrotic material in APAP-ALI since host KCs are depleted. AAMs accounted for 0.9 % of the liver myeloid population 20 hours post-transfer in APAP-ALI mice. AAMs were highly phagocytic both in culture and in liver tissue (80 % Ly6C<sup>10</sup>, > 99 % PKH-uptake). Furthermore, phagocytosis was poor (8.2 % PKH-uptake) in endogenous Ly6C<sup>hi</sup> infiltrating macrophages in APAP-ALI mice but improved with AAM-treatment (12 % PKH uptake). This suggests that AAM-treatment can modulate the dynamics of the host immune response amplifying the therapeutic effect. Consistent with this, AAMs were ineffective when delivered at four hours post-APAP in the high-dose APAP (500 mg/kg) experiment, i.e. before CCR2+ monocytes infiltrate the liver.<sup>[32]</sup> Indeed, hAAMs were ineffective in immunocompromised mice, suggesting an immunocompetent system is required to achieve

efficacy. In immunocompetent mice, hAAMs improved necrosis resolution and hepatocyte proliferation (albeit requiring higher number of transferred cells). However, hAAMs had no effect on serum proinflammatory cytokine levels, possibly due to rapid clearance of human cells by host NK cells, or the lack of host response from any hAAM-derived factors due to species differences in immune signalling pathways. AAM efficacy is therefore likely to be attributed to both the phagocytosis of necrotic tissue and paracrine effects upon the host immune system. Mouse and human AAMs took eight days to generate using current protocols - a timeframe that

is clinically incompatible with the emergency settings of an APAP overdose. A potential clinical therapeutic product would therefore likely be allogenic, immunocompatible, scalable, and permit cryopreservation so that cells could be rapidly administration in an emergency scenario. In summary, AAM therapy for ALI reduces necrosis and inflammation and increases liver regeneration and is a potential novel therapy for late presenting APAP-ALI.

Acknowledgments: We thank W. Mungall and F. Howie for help with animal experiments and analytical support respectively.

# REFERENCES

[1] Larson AM, Polson J, Fontana RJ, Davern TJ, Lalani E, Hynan LS, et al. Acetaminophen-induced acute liver failure: Results of a United States multicenter, prospective study. Hepatology 2005;42:1364.

[2] Lee WM. Acetaminophen and the U.S. acute liver failure study group: Lowering the risks of hepatic failure. Hepatology 2004;40:6.

[3] Lee WM. Acetaminophen-related acute liver failure in the United States. Hepatology Research 2008;38:S3.
 [4] Prescott LF, Illingworth RN, Critchley JAJH. Intravenous N-acetylcysteine: The treatment of choice for paracetamol poisoning. BrMedJ 1979;2:1097.

[5] Rolando N, Wade J, Davalos M, Wendon J, Philpott-Howard J, Williams R. The systemic inflammatory response syndrome in acute liver failure. Hepatology 2000;32:734-739.

[6] Zigmond E, Samia-Grinberg S, Pasmanik-Chor M, Brazowski E, Shibolet O, Halpern Z, et al. Infiltrating monocyte-derived macrophages and resident kupffer cells display different ontogeny and functions in acute liver injury. J Immunol 2014;193:344-353.

[7] Gao B, Jeong WI, Tian Z. Liver: An organ with predominant innate immunity. Hepatology 2008;47:729-736.

[8] Canalese J, Gove CD, Gimson AE, Wilkinson SP, Wardle EN, Williams R. Reticuloendothelial system and hepatocytic function in fulminant hepatic failure. Gut 1982;23:265-269.

[9] Rolando N, Harvey F, Brahm J, Philpott-Howard J, Alexander G, Gimson A, et al. Prospective study of bacterial infection in acute liver failure: an analysis of fifty patients. Hepatology 1990;11:49-53.

[10] Rolando N, Philpott-Howard J, Williams R. Bacterial and fungal infection in acute liver failure. Semin Liver Dis 1996;16:389-402.

[11] Laskin DL, Gardner CR, Price VF, Jollow DJ. Modulation of macrophage functioning abrogates the acute hepatotoxicity of acetaminophen. Hepatology 1995;21:1045.

[12] Goldin RD, Ratnayaka ID, Breach CS, Brown IN, Wickramasinghe SN. Role of macrophages in acetaminophen (paracetamol)-induced hepatotoxicity. J Pathol 1996;179:432-435.

[13] Michael SL, Pumford NR, Mayeux PR, Niesman MR, Hinson JA. Pretreatment of mice with macrophage inactivators decreases acetaminophen hepatotoxicity and the formation of reactive oxygen and nitrogen species. Hepatology 1999;30:186.

[14] You Q, Holt M, Yin H, Li G, Hu CJ, Ju C. Role of hepatic resident and infiltrating macrophages in liver repair after acute injury. Biochem Pharmacol 2013;86:836-843.

[15] Ju C, Reilly TP, Bourdi M, Radonovich MF, Brady JN, George JW, et al. Protective role of Kupffer cells in acetaminophen-induced hepatic injury in mice. Chem Res Toxicol 2002;15:1504-1513.

[16] Holt MP, Yin H, Ju C. Exacerbation of acetaminophen-induced disturbances of liver sinusoidal endothelial cells in the absence of Kupffer cells in mice. Toxicol Lett 2010;194:34-41.

[17] Triantafyllou E, Pop OT, Possamai LA, Wilhelm A, Liaskou E, Singanayagam A, et al. MerTK expressing hepatic macrophages promote the resolution of inflammation in acute liver failure. Gut 2018;67:333-347.

[18] Brouckaert G, Kalai M, Krysko DV, Saelens X, Vercammen D, Ndlovu MN, et al. Phagocytosis of necrotic cells by macrophages is phosphatidylserine dependent and does not induce inflammatory cytokine production. Mol Biol Cell 2004;15:1089-1100.

[19] Cocco RE, Ucker DS. Distinct modes of macrophage recognition for apoptotic and necrotic cells are not specified exclusively by phosphatidylserine exposure. Mol Biol Cell 2001;12:919-930.

[20] Thomas JA, Pope C, Wojtacha D, Robson AJ, Gordon-Walker TT, Hartland S, et al. Macrophage therapy for murine liver fibrosis recruits host effector cells improving fibrosis, regeneration, and function. Hepatology 2011;53:2003-2015.

[21] Moore JK, Mackinnon AC, Wojtacha D, Pope C, Fraser AR, Burgoyne P, et al. Phenotypic and functional characterization of macrophages with therapeutic potential generated from human cirrhotic monocytes in a cohort study. Cytotherapy 2015;17:1604-1616.

[22] Fraser AR, Pass C, Burgoyne P, Atkinson A, Bailey L, Laurie A, et al. Development, functional characterization and validation of methodology for GMP-compliant manufacture of phagocytic macrophages: A novel cellular therapeutic for liver cirrhosis. Cytotherapy 2017.

[23] Moroni F, Dwyer BJ, Graham C, Pass C, Bailey L, Ritchie L, et al. Safety profile of autologous macrophage therapy for liver cirrhosis. Nat Med 2019;25:1560-1565.

[24] Dufour DR, Lott JA, Nolte FS, Gretch DR, Koff RS, Seeff LB. Diagnosis and monitoring of hepatic injury. I. Performance characteristics of laboratory tests. Clin Chem 2000;46:2027-2049.

[25] Liu ZX, Han D, Gunawan B, Kaplowitz N. Neutrophil depletion protects against murine acetaminophen hepatotoxicity. Hepatology 2006;43:1220-1230.

[26] Jaeschke H. How relevant are neutrophils for acetaminophen hepatotoxicity? Hepatology 2006;43:1191-1194.

[27] Jaeschke H, Liu J. Neutrophil depletion protects against murine acetaminophen hepatotoxicity: another perspective. Hepatology 2007;45:1588-1589; author reply 1589.

[28] Yang W, Tao Y, Wu Y, Zhao X, Ye W, Zhao D, et al. Neutrophils promote the development of reparative macrophages mediated by ROS to orchestrate liver repair. Nat Commun 2019;10:1076.

[29] Silva MT. Macrophage phagocytosis of neutrophils at inflammatory/infectious foci: a cooperative mechanism in the control of infection and infectious inflammation. J Leukoc Biol 2011;89:675-683.

[30] Lu B, Antoine DJ, Kwan K, Lundback P, Wahamaa H, Schierbeck H, et al. JAK/STAT1 signaling

promotes HMGB1 hyperacetylation and nuclear translocation. Proc Natl Acad Sci U S A 2014;111:3068-3073.

[31] Ramachandran P, Pellicoro A, Vernon MA, Boulter L, Aucott RL, Ali A, et al. Differential Ly-6C expression identifies the recruited macrophage phenotype, which orchestrates the regression of murine liver fibrosis. Proc Natl Acad Sci U S A 2012;109:E3186-3195.

[32] Mossanen JC, Krenkel O, Ergen C, Govaere O, Liepelt A, Puengel T, et al. Chemokine (C-C motif) receptor 2-positive monocytes aggravate the early phase of acetaminophen-induced acute liver injury. Hepatology 2016.

[33] Holt MP, Cheng L, Ju C. Identification and characterization of infiltrating macrophages in acetaminopheninduced liver injury. J Leukoc Biol 2008;84:1410-1421.

[34] Nishida M, Okumura Y, Fujimoto S, Shiraishi I, Itoi T, Hamaoka K. Adoptive transfer of macrophages ameliorates renal fibrosis in mice. Biochem Biophys Res Commun 2005;332:11-16.

[35] Leung G, Petri B, Reyes JL, Wang A, Iannuzzi J, McKay DM. Cryopreserved Interleukin-4-Treated Macrophages Attenuate Murine Colitis in an Integrin beta7 - Dependent Manner. Mol Med 2016;21:924-936.
[36] Scaffidi P, Misteli T, Bianchi ME. Release of chromatin protein HMGB1 by necrotic cells triggers inflammation. Nature 2002;418:191.

[37] Michalopoulos GK, DeFrances MC. Liver regeneration. Science 1997;276:60-66.

[38] Ding BS, Cao Z, Lis R, Nolan DJ, Guo P, Simons M, et al. Divergent angiocrine signals from vascular niche balance liver regeneration and fibrosis. Nature 2014;505:97-102.

[39] Kato T, Ito Y, Hosono K, Suzuki T, Tamaki H, Minamino T, et al. Vascular endothelial growth factor receptor-1 signaling promotes liver repair through restoration of liver microvasculature after acetaminophen hepatotoxicity. Toxicol Sci 2011;120:218-229.

Journal Press

# **FIGURE LEGENDS**

Fig. 1. Injection of AAMs reduces necrosis and stimulates liver regeneration following APAP-ALI. (A) Four macrophage populations, derived from mouse BM, were generated for testing: 1. unstimulated BMDMs (Naïve), 2. CAMs, 3. AAMs, and 4. DAMs. (B) Study design: injection of macrophages (1x10<sup>6</sup>, i.v.) or PBS alone into APAP-ALI mice at 16 hours, before cull at 36 hours. (C) Serum ALT activity (left panel) and AST activity (right panel) in APAP-ALI mice receiving indicated treatments (D) Representative liver histological stains from APAP-ALI mice receiving indicated treatments, necrosis quantification in right panel (E) Representative Ly6G IHC stains in liver from APAP-ALI mice with indicated treatments. Black arrows indicate Ly6G-positive cells, quantification in right panel. (F) Representative images of HMGB1 IHC stains of liver tissue from APAP-ALI mice with indicated treatments. Black arrows indicate HMGB1-negative nuclei, quantification in right panel. (G) Representative IF stains of BrdU incorporation (yellow nuclei, indicated by white arrows) in liver with DAPI counterstain (cyan) from APAP-ALI mice with indicated treatments. Quantification in right panel. (H/I) Dual IF stains of BrdU (yellow), and either HNF4a (H) or ERG (I) (magenta), with DAPI counterstain (cyan) in AAM-treated liver tissue. White arrows indicate dual-positive cells, quantification in right panel. All data shown are n = 6-12 mice per group (black circles – individual vehicle controls; grey circles - cell-transfer reference group; open circles - polarised-BMDM treated mice). Scale bars - 100 µm. p-values indicated in panels, n.s. – not significant. Kruskal-Wallis test for (C/D/E/F/G/H). One-way ANOVA for I.

Fig. 2. Injection of AAMs reduces several serum proinflammatory cytokines following APAP-ALI. (A) Serum concentrations of proinflammatory cytokines measured in APAP-ALI mice receiving indicated treatments (n = 10-12 per group; some sera had undetectable IL-12p70). (B) IL-6 levels in liver homogenates from APAP-ALI mice receiving indicated treatments (n =<u>5-11 per group</u>) (C) Relative expression of indicated genes (using  $2^{-\Delta\Delta CT}$  method; standardised to PBS-treated controls, after GAPDH normalisation) in liver tissue of APAP-ALI mice receiving indicated treatments (n = 6-10 per group). In A-C, black circles – individual vehicle controls; grey circles - cell-transfer reference group; open circles - AAM-treated mice. (D) Study design: Plasma biomarkers were measured daily in APAP-ALI mice following PBS/AAM-treatment. (E/F/G) Plasma biomarkers (left panels) and change in plasma biomarkers from poinf-oftreatment (right panels) for plasma ALT activity (E), AST activity (F), and miR-122 levels (G). Amber area represents treatment phase. Grouped values represents mean ± SD for AAMtreatment (green) and PBS-treatment (black). Plasma miR-122 levels are presented as relative quantitation (using  $2^{-\Delta\Delta CT}$  method; standardized to pre-APAP-ALI levels (-96 hours), after let-7d normalization). *p*-values provided in panels; n.s., not significant, \**p*<0.05, \*\**p*<0.01. Kruskal-Wallis tests for A/B/C (Csf1/Ccl5), one-way ANOVA for C (Tgfb, Il6, Cxcl1), mixed-effects model for (E/F/G).

Fig. 3. Injected AAMs are primarily Ly6C<sup>lo</sup> and highly phagocytic *in situ*. (A) Relative gene expression in BMDM populations (determined by  $2^{-\Delta\Delta CT}$  method; standardised to Naive BMDMs, after 18S normalisation): Nos2 - CAM-associated gene, Retnla - AAM-associated gene, <u>ll10 - DAM-associated gene, Ly6C</u> - proinflammatory-associated gene (n = 3/4 biological replicates per group). (B) Ly6C status in BMDM populations (flow cytometry quantification, left panel; cytometry histograms, right panel) (C) Study design: CFSE-labelled AAMs injected  $(5 \times 10^6 \text{ i.v.})$  at 16 hours in APAP-ALI mice, three hours before PKH26 (fluorescent phagocytic tracer). Cull at 36 hours. (D) Serum ALT activity (left) and AST (right) in APAP-ALI mice treated with PBS or AAMs (n=4 per group). (E) Gating shows CFSE+ AAMs in liver digests (top panels) and whole blood (bottom panels) in AAM-treated APAP-ALI mice. (F). Representative flow plot of Ly6C status in retrieved CFSE+ AAMs (gating: left panel; quantification: right panel, circles represent digests from individual mice). (G) Representative flow plots showing PKH-uptake in Ly6C<sup>hi</sup> (left panel) and Ly6C<sup>lo</sup> (middle panel) AAMs, quantification in right panel. (H) Representative flow plots show Ly6C status in infiltrating endogenous macrophages in liver digests from APAP-ALI mice treated with PBS (left panel) or AAMs (middle panel), quantification in right panel. (I) Quantification of PKH-uptake in Ly6Chi (left panel) and Ly6C<sup>lo</sup> (right panel) infiltrating endogenous macrophages. *p*-values provided in panels, n.s. not significant. One-way ANOVA (A), unpaired t-test (D, I), or Mann-Whitney Utest (H) were performed.

Fig. 4. Murine AAMs are highly phagocytic. (A) Phagocytosis quantification in naive BMDMs (black), AAMs (green), or CAMs (blue) during incubation with CMTMR-labelled apoptotic thymocytes. MFI of CMTMR-positive macrophage populations (left), and representative histograms (right). (B) Ly6C status in BMDM populations during incubation with CMTMR-labelled apoptotic thymocytes at indicated times (left panel); representative histograms (right panel). Coloured circles represent individual preparations connected by lines (mean value, n=3). (C) Representative images of real-time phagocytosis at indicated times. Naïve BMDMs (top row), CAMs (middle row) and AAMs (bottom row) are shown (Deep Red CellMask, red; NucBlue, blue). Phagocytosis determined by intracellular fluorescence (green). (D) Phagocytosis quantification: pHrodo-positive cell fraction (left) and total cell MFI (right). (E) Study design: Hepatocytes were Tdtomato-labelled by delivery of hepatotropic AAV8 virus delivering Crerecombinase to R26RLSLtdTomato mice. Tdtomato-positive APAP-ALI mice received CFSElabelled AAMs (5x10<sup>6</sup>, i.v.) at 16 hours, before cull (36 hours). (F) Panels show representative confocal immunofluorescence of liver tissue (max intensity projection from 7 slices; 2.4 µm) in each channel: DAPI (cyan), TdTom (TdTomato+ hepatocytes, magenta), FITC (CFSE+ AAMs, yellow), and merged images. Faint punctate TdTomato+ debris were visible inside vesicles in peri-necrotic macrophages (top row, white arrow heads). \*p < 0.05, \*\*p < 0.01, \*\*\*p < 0.001, scale bars 20 µm. Two-way ANOVA (A/B).

Fig. 5. Injection of hAAMs reduces necrosis in APAP-ALI immunocompetent mice. (A) hMDMs were differentiated from CD14+ cells isolated from healthy volunteer buffy coats, before incubation with hCSF-1 for seven days. hAAMs were generated by stimulating hMDMs overnight with hCSF1, hIL-4, and hIL-13. (B) Representative flow cytometry plots demonstrating CD14-enrichment using CliniMACS® beads (C) Representative flow cytometry plots showing macrophage maturity markers (25F9, left panels; CD206, right panels) in CD14+ cells (top) and hMDMs (bottom) (D) Flow cytometry quantification of hMDMs (black), hMDMs stimulated with CSF-1 alone (grey), and fully-stimulated hAAMs (green), spots represent individual donors. (E) Panels show representative anti-FITC IHC stains in liver (left), spleen (centre), and lung (right) from APAP-ALI mice treated with PBS (top row) or hAAMs (bottom row). (F) Representative histological staining (left) and necrosis segmentation map (centre) in APAP-ALI mice with indicated treatments. Necrosis quantification in right panel ( $n \ge 7$  mice per group). (G) Percentage weight loss (left panel) and liver/body weight ratio (right panel) of APAP-ALI mice with indicated treatments ( $n \ge 8$  per group) (H). Serum injury/inflammatory markers in APAP-ALI mice receiving indicated treatments ( $n \ge 6$  per group; serum ALT activity, left panel; serum IL-6, centre panel; serum IL-10/-12p70 ratio, right panel, some sera has undetectable IL-12p70 levels). (I) Representative dual IF images of HNF4a (yellow) and BrdU (magenta) against DAPI (counterstain, blue) in liver tissue from APAP-ALI mice treated with PBS (top row) or hAAMs (bottom row). White arrowheads indicate BrdU-positive cells, red arrowheads indicate dual-positive BrdU-positive HNF4 $\alpha$ -positive cells (quantification right panel;  $n \ge 8$ ). Scale bars 100 µm, unless otherwise indicated. *p*-values indicated in panels, n.s. not significant. One-way ANOVA test in F, G (weight loss %), H (ALT), or Kruskal-wallis test in G (liver/body ratio), H (IL-6, IL-10/-12 ratio), and I.









R26RLSLtdTomato mice



Journal Pre-proof Journal Pre-proof



# **Highlights:**

- Primary BMDMs localised in liver and spleen within hours following intravenous injection in mice
- AAMs were highly phagocytic and, after injection, were most effective at reducing necrotic area, HMGB1 translocation, and hepatic neutrophil infiltration following APAP-induced liver injury
- AAM injection reduced several inflammatory mediators in both serum and tissue, and stimulated hepatocyte and endothelium proliferation in injured liver
- Injection of clinical-grade human AAMs could partially recapitulate the efficacy of murine AAMs in immunocompetent mice

at the second